Author's response to reviews

Title: An evaluation of inflammatory gene polymorphisms in sibships discordant for premature coronary artery disease: The GRACE-IMMUNE study

Authors:

Benjamin D Brown (benbrown@doctors.org.uk)
Jeremie Nsengimana (J.Nsengimana@leeds.ac.uk)
Jennifer H Barrett (J.H.Barrett@leeds.ac.uk)
Richard A Lawrence (richardlawrance@hotmail.com)
Lori Steiner (lori.steiner@roche.com)
Suzanne Cheng (suzanne.cheng@roche.com)
D T Bishop (D.T.Bishop@leeds.ac.uk)
Nilesh J Samani (njs@leicester.ac.uk)
Stephen G Ball (S.G.Ball@leeds.ac.uk)
Anthony J Balmforth (A.J.Balmforth@leeds.ac.uk)
Alistair S Hall (cvsash@leeds.ac.uk)

Version: 2 Date: 1 December 2009

Author's response to reviews:

N/A